Author’s response to reviews

Title: HYPERHOMOCYSTEINEMIA IN BILATERAL ANTERIOR ISCHEMIC OPTIC NEUROPATHY AFTER CONVENTIONAL CORONARY ARTERY BYPASS GRAFT: A CASE REPORT

Authors:
Alfredo Niro (alfred.nir@tiscali.it)
Giancarlo Sborgia (gcsborgia@gmail.com)
Alessandra Sborgia (alessandrasborgia@yahoo.it)
Giovanni Alessio (giovanni.alessio@uniba.it)

Version: 3 Date: 09 Aug 2017

Author’s response to reviews:

Dear Editor,

Please find herewith enclosed the revised version of our manuscript entitled “HYPERHOMOCYSTEINEMIA IN BILATERAL ANTERIOR ISCHEMIC OPTIC NEUROPATHY AFTER CONVENTIONAL CORONARY ARTERY BYPASS GRAFT: A CASE REPORT” for publication on your Journal.

Find below the answer to the Reviewer(s) comments:

The text has been revised by a native English speaker.

We therefore hope that the manuscript now meets the interest of the editor and merits publication in your Journal.

Yours faithfully,
Alfredo Niro, PhD